Author's response to reviews

Title: A de novo complete BRCA1 gene deletion identified in a Spanish woman with early bilateral breast cancer.

Authors:

Zaida Garcia-Casado (zaida.garcia@uv.es)
Ignacio Romero (nachete54@hotmail.com)
Antonio Fernandez-Serra (anfersefar@yahoo.es)
Luis Rubio (rubio_lui@hotmail.com)
Francisco Llopis (fsllopis@hotmail.com)
Ana Garcia (anagarcialloret@yahoo.es)
Pilar Llombart (pllombart@correo.cop.es)
Jose A. Lopez-Guerrero (jalopez@fivo.org)

Version: 5 Date: 10 October 2011

Author's response to reviews: see over
Dear Editors,

Please find enclosed the original manuscript of the article prepared by our group entitled “A de novo complete BRCA1 gene deletion identified in a Spanish woman with early bilateral breast cancer” and referenced as 7689819775421449 by your editorial office.

Sincerely yours,

Dr. José Antonio López-Guerrero
Corresponding author